
diff -r 942205605c30 -r 34a8f73b2c94 handouts/ho01.tex
--- a/handouts/ho01.tex	Tue Dec 30 21:59:47 2014 +0000
+++ b/handouts/ho01.tex	Wed Dec 31 01:49:20 2014 +0000
@@ -4,10 +4,13 @@
 
 \lstset{language=JavaScript}
 
+
 \begin{document}
+\fnote{\copyright{} Christian Urban, 2014}
 
 \section*{Handout 1 (Security Engineering)}
 
+
 Much of the material and inspiration in this module is taken
 from the works of Bruce Schneier, Ross Anderson and Alex
 Halderman. I think they are the world experts in the area of




